Author’s response to reviews

Title: Functional examination of MLH1, MSH2, and MSH6 intronic mutations identified in Danish colorectal cancer patients

Authors:

Sanne M. Petersen (sannemiepetersen@gmail.com)
Mette Dandanell (mettedandanell@gmail.com)
Lene J. Rasmussen (lenera@sund.ku.dk)
Anne-Marie Gerdes (Anne-Marie.Gerdes@regionh.dk)
Lotte N. Krogh (Lotte.Krogh@ouh.regionsyddanmark.dk)
Inge Bernstein (i.bernstein@rn.dk)
Henrik Okkels (heok@rn.dk)
Friedrik Wikman (FPW@KI.AU.DK)
Finn C. Nielsen (tcn@rh.dk)
Thomas v.O. Hansen (tvoh@rh.dk)

Version: 2 Date: 22 March 2013

Author’s response to reviews:

We have now stated in the methods section (patients and clinical data) that written informed consent was obtained from all probands (all adults). It has also been stated in this section that since the study is part of normal diagnostic procedures no ethical approval was obtained, and that the study was conducted in accordance with the Helsinki Declaration.